Author's response to reviews

Title: Women's attitudes towards mechanisms of action of family planning methods: survey in primary health centres.

Authors:

Jokin de Irala (jdeirala@unav.es)
Cristina Lopez del Burgo (cldelburgo@unav.es)
Carmen M Lopez de Fez (cdefez@yahoo.es)
Jorge Arredondo (djarrecha@yahoo.es)
Rafael T Mikolajczyk (rafael.mikolajczyk@uni-bielefeld.de)
Joseph B Stanford (jstanford@dfpm.utah.edu)

Version: 2 Date: 5 February 2007

Author's response to reviews:

Deborah Saltman, MD, PhD

BMC Women's Health

BioMed Central Ltd,

Middlesex House, 34-42 Cleveland Street

London W1T 4LB, UK.

Dear Mrs. Saltman, editor,

Please find enclosed the manuscript entitled "Women's attitudes towards mechanisms of action of family planning methods: survey in primary health centres", which we would like to be considered for publication in BMC Women's Health. The manuscript represents original work not submitted anywhere else.

The manuscript conforms to the Instructions for Authors available from the Journal's Website. All authors have agreed to submit this article in the present form, and are prepared to take public responsibility for its content. Each has participated sufficiently in the conception and design of the work or in the analysis of the data. All authors have fully participated in the writing and editing of the manuscript. The final manuscript has been seen and approved by all authors and they have taken due care to ensure the integrity of the work. Each author attests that he will produce the data upon which the manuscript is based for examination by the editors or their assignees if requested.

We enclose the spanish questionnaire used for this study.

We look forward to your comments and decision on our contribution in due time.

Sincerely,

Dr. Cristina Lopez del Burgo